RICH1 Meeting Friday 22nd Feb 2008

Present

CERN: 
Jon Imong, Fabio Metlica, Richard Plackett, Olav Ullaland.
Imperial: 
Bill Cameron, Dave Websdale.
Oxford: 
Jim Libby, Neville Harnew 

4) Lower HPD Plane
4.1/4.2) Box Manufacture
· All of the box parts that were sent for external water-cutting will be collected this afternoon.

· These and all the other available parts for the box will be sent to be anodized on Monday morning.

· Assuming a nominal week for the work to be done the parts will be back on Monday 3rd of March.

· From experience it will take the rest of that week to assemble the parts to make the box and to check for errors.

· We should be able to set an optimistic date for the delivery of the second box no earlier than Friday 7th March.

· Non-anodized parts will be made in the Imperial workshop whilst the other parts are away being anodized. These include the wheel support blocks and other external pieces.

· A new data end plate is currently being made and will be sent to be anodized on Monday. This has an 18mm offset for the data patch panels to reduce stress on the cables inside the box.

4.3/4.4) Installation Frame 
· The rail system and hoist are currently in the Oxford workshop.

· It is expected that the full system will take three to four weeks to complete. This timescale moves delivery of the system to the week ending 21st of March. 
· To speed things up as much as possible some guidance on priorities was requested by Jim.
· The rails to fit inside the shield and their supports are the highest priority and will be taken to CERN by car as soon as they are ready to check their compatibility with the shield.

· The hoist and its frame is not a high priority as we can probably do this operation by crane. 
· The column extraction frame is also a low priority as it isn’t required for the initial installation phase of the work.

· It was noted that Rohan had come back a week early after his operation and we all offer our thanks for his hard work.

· Bill and Neville will inform Werner and Rolf of the lower box schedule changes. 
1) Magnetic Shield
1.1) Securing Magnetic Shield
· All the necessary pieces of the clamps to secure the lower magnetic shield against the force of the dipole magnet have now been fabricated.

· The steel plates that will take the clamps have been fitted to the concrete floor beneath the lower shield.

· The installation of these clamps is scheduled to happen on Monday and Tuesday next week (25/2/08).

1.2) Survey Plans
We need to re-survey the position of the gas enclosure as the lower shield may have moved in the magnet test in October. At the same time the position of the upper HPD box will be measured. This has to happen after the clamps are installed and is scheduled for Thursday of next week. This survey will use 4 points on each side of the gas enclosure and two points on the HPD box.  All these points are currently visible from the standard survey mounts.

We will take at least a day or two to examine the data from the survey to decide on whether and how the shield/gas enclosure/HPD box have to be moved.

2) Gas Enclosure

2.1) Machining the Feet
In order for the current design of the lower photon funnel to fit below the gas enclosure the clamps that hold its upstream feet need to be modified.  This has been done and they were replaced this morning in time for the shield clamping next week.

2.2) Patch Panels
The patch panels for the gas enclosure have yet to be designed in detail.  Currently it is foreseen that they will hold the temperature and humidity monitoring for the gas volume (which is already in place) and the LAMS system fibre and camera cable feed-throughs.  The cable feed-throughs for the LAMS cameras will have to be a piece potted around a cable. These cables cannot be broken as they include a tuned transmission line due to the design of the cameras.

2.3) Gas System
· Some machining of the proposed connectors is required.

· The bubblers that we plan to use are available and being tested for leak tightness in Erich’s lab.

· The flexible tube to connect between the rigid pipes in the hall and the RICH1 GE has to be ordered, but we now know what length it needs to be.
· There are some concerns around the removal of the HPD box with the pipe in place but a low profile connector can probably be prepared.

3) Upper HPD Box Status
3.1) Ladder Installation and Testing
· The ladder installation and testing was successfully completed by Feb 15th.  

· Several problems were encountered with column M2 which had to be removed and replaced twice.  The first time because the SPECS mezzanine board had become detached due to the arrangement of the cables on the column, and the second time because a short circuit between the temperature monitoring system and the JTAG data line was causing the column to fail its electrical checks. Thanks to Ken for resolving this.
· From the installation work two changes are proposed for the lower box.  The first is the off-setting of the data patch panels to relieve some pressure on the SPECS cable and TTC fibres and the second is replacing the ‘U’ mounting blocks with insulating versions to aid the electrical isolation of the column.

3.2) Installation and Safety Issues

· The box was successfully installed on the morning of 18th Feb.

· There were concerns regarding the lifting beam that meant we were only allowed to use it in its present configuration once.  The use of threaded rod for lifting is not allowed, and any modifications to a lifting beam means it has to be recertified before it can be used.  The redesign is a matter of some urgency as the lifting gear is needed to perform the fine position adjustment of the box.
3.3) Position Adjustment and Survey 

· A 5mm downstream shift in the direction parallel to the edge of the quartz window has been achieved by shimming the mounts for the wheels of the upper box. This leaves us with only 2mm clearance from the downstream iron wall.  

· The height from the plinth on the HPD box to the sealing surface on the gas enclosures (measured at the corners) is ~121mm. Christophe’s design calls for a gap of 126mm. This means that the maximal shift of 5mm upwards has to be achieved but is prevented right now by the clearance with the iron unless the lower shielding is moved. A decision on the best way to proceed will be taken only after the survey next week.

·  To adjust the height of the upper box the crane will be required, raising the question of the safety of the lifting beam (see 3.2 above).

3.4) Cabling Status

· The wooden template for the door has been used to define the exact position of the cable port in the shielding doors through which the cables must be routed.
A scaffold was erected to permit easy access to the top of RICH1.
· All the cables on the balcony are connected with the exception of the DSS cables.

· All the cables at the RICH1 upper box are crimped except the UHV connectors, two SHV connectors for the Light-Tight PMTs and a single DSS 37pin Sub-D connector.

· The following cables have been tested: UHV, T-mon, HV-mon.

· The ELMB system that monitors the temperature and pressure in the RICH1 upper box has been read out with the PVSS control software.  The humidity is currently ~10%

3.5) Cooling Pipe Routing and Connection
There is a possible delay with the connection of the cooling pipes to the manifold. A length of the SS flexile pipe has to be ordered; the connectors will be the same as those used in RICH2.  This flexible link needs to be of the order of 3m long.
3.6) Cooling Manifold
The cooling manifold has yet to be produced. It is possible for us to use a temporary cooling manifold to begin to commission the system.

3.7) Position End Stops
The chariot is currently clamped with ‘G’ clamps to constrain its position and secure it on the rails.  The precision end stops have yet to be produced.  
3.8) UHV Safety Interlock
This is still pending. The wooden door template will be very useful in designing this.

5) Photon Funnels
· The quartz frames are currently in manufacture at a company in Italy.

· The photon funnel design has been finalized and been sent to PSI for manufacture.

· The quartz pane gluing is complete with the two halves of both panes glued and left for curing.

· The GE feet modifications to allow the installation of the bottom PF are complete

· The lower HPD box plinth may need to be modified. Drawings are required from Christophe before the box leaves the Imperial workshop.
· Christophe is currently working on the design for the flexible coupling to link the PF and the HPD box.

6) Magnetic Distortion Monitoring System 
This now has a dedicated weekly meeting on Tuesdays, please look at its minutes /slides to see the progress.

7) Mirrors: Dust removal plans
Awaiting the next opening of the gas enclosure.

8) LAMS
No progress since last meeting.


9) Cables still to be installed 
HV interlock, LAMS (camera and fibre), MDMS lower data and TTC fibres, lower LT PMT cables.  (Action RP to prepare definitive list).
Minutes by Richard Plackett
